Author's response to reviews

Title: The burden and treatment of HIV in tuberculosis patients in Papua Province, Indonesia: a prospective observational study

Authors:

Gysje J Pontororing (gysje_jp@yahoo.com)
Enny Kenangalem (enny_timika@yahoo.co.id)
Dina B Lolong (marvin_lolong@yahoo.com)
Govert Waramori (Govert_Waramori@fmi.com)
. Sandjaja (san_gizi@yahoo.com)
Emiliana Tjitra (emil@jakarta.wasantara.net.id)
Ric N Price (ric.price@menzies.edu.au)
Paul M Kelly (paul.kelly@anu.eu.au)
Nicholas M Anstey (Nicholas.Anstey@menzies.edu.au)
Anna P Ralph (anna.ralph@menzies.edu.au)

Version: 4 Date: 21 December 2010

Author's response to reviews:

The final proofed version of the manuscript (0.corrected.doc), Figure 1 (1.corrected.doc) and Figure 2 (2.corrected.doc) are now attached.